

REVIEW HISTORY

Paper: “Developing Student Growth Through Effective Inclusion Skill Sets in the Rural Black Belt Region of Alabama and Mississippi”

Corresponding Author: Kimberly Grantham Griffith
Email: kgriffith@uwa.edu

Doi: 10.19044/ejes.v6no3a2

Peer review:
Reviewer 1: Blinded
Reviewer 2: Blinded

Published: 31.09.2019